Author’s response to reviews

Title: Mediastinitis complicating a percutaneous endoscopic gastrostomy: a case report

Authors:

Kalliopi K Papakonstantinou (akarab@forthnet.gr)
Dr Athanasios Karagiannis (cava@otenet.gr)
Dr Maria J Tsirantonaki (saitanis@aua.gr)
Anastasios Konstantinidis (akarab@forthnet.gr)
Spiros Spirou (akarab@forthnet.gr)
Ion Skottis (akarab@forthnet.gr)
Andreas P Karabinis (akarab@forthnet.gr)

Version: 3 Date: 5 May 2003

COMMENTS ON THE SUGGESTIONS OF THE REVIEWERS

We thank reviewers for their positive comments.

Dr. Callum B. Pearce suggested no specific alterations although he thinks that the table would not have been included since it contains information reported elsewhere.

This table, rich of up to date references, has been constructed, after reviewing relevant literature concerning PEG complications, to emphasize the rarity of mediastinitis associated with PEG. So, we think it is better the table to be included in this case report.

Dr. Ioannis Koutroubakis suggests the inclusion of the CT scan pictorial findings.

We now include the relevant CT scan (Figure 1a,b).
The legend of the Figure 1a,b has been added in a separate page at the end of the ms.